Author's response to reviews

**Title:** Knowledge, attitudes and practices regarding gemstone therapeutics in a selected adult population in Pakistan

**Authors:**

Sidra Ishaque (<sidraishaque@gmail.com>)
Taimur Saleem (<taimur@gmail.com>)
Waris Qidwai (<waris.qidwai@aku.edu>)

**Version:** 11  **Date:** 16 August 2009

**Author's response to reviews:** see over
Title: Knowledge, attitudes and practices regarding gemstone therapeutics in a selected adult population in Pakistan

Authors: Sidra Ishaque, Taimur Saleem, Waris Qidwai

We are extremely grateful to the reviewer(s) for pointing out these shortcomings in the manuscript. We have rectified them to the best of our ability and hope that they are up to the reviewer’s satisfaction. We feel privileged that our manuscript is being published in BioMed Central.

Thanks and Regards,

Authors
Responses to Reviewer’s Report:

Major revisions (We require the author to make these changes)

Reference citations - please cite all references in numerical order. Reference 22 is currently not cite.

Response: This has been done. All the references have been checked.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above)

1. Underlining - please remove from the manuscript. You may replace it with bold if you wish.

Response: Underlining has been removed from the manuscript.

2. List of abbreviations - please rename Abbreviations and format as a paragraph rather than a list.

Response: This has been done as suggested. The abbreviations are in a paragraph form and the heading has also been renamed.

3. Tables - please remove vertical lines from the tables as they are not supported in the final text.

Response: This change has been made in all the tables.

4. Page 38 - please remove this blank page.

Response: This has been done.